
\documentstyle[11pt]{article}
\textwidth=6.5in
\textheight=9in
\evensidemargin=0in
\oddsidemargin=0in
\topmargin=-0.3in
\topskip=0pt
\baselineskip=12pt
\parskip=0pt 
\parindent=0em
\begin{document}


\section{Presentation Information}
        \begin{itemize}
        \item Your Name: \hrulefill .
        \item Student: \hrulefill .
        \item Paper Title: \hrulefill .
        \item Authors:  \hrulefill .
        \item Citation Information: \hrulefill .
        \end{itemize}

\section{Evaluation}

	Please rank these on a scale of 1 to 5,  where 1 is poor and 5
	is excellent.

	\begin{itemize}
	\item Presentation
		\begin{itemize}
		\item \underline{\hspace{0.5in}}Preparation
		\item \underline{\hspace{0.5in}}Understanding of material
		\item \underline{\hspace{0.5in}}Command of the English
			Language (Clarity, vocabulary, \ldots)
		\item \underline{\hspace{0.5in}}Time Limit.
		\end{itemize}
	\item Content
		\begin{itemize}
		\item \underline{\hspace{0.5in}}Was the introduction to
			the problem interesting? (provide motivation?)
		\item \underline{\hspace{0.5in}}Was material presented
			in a fluent order/manner?
		\item \underline{\hspace{0.5in}}Did the speaker assume
			too much knowledge?
		\end{itemize}
	\item Examples
		\begin{itemize}
		\item \underline{\hspace{0.5in}}Enough
		\item \underline{\hspace{0.5in}}Clear and Concise
		\end{itemize}
	\item Figures
		\begin{itemize}
		\item \underline{\hspace{0.5in}}Clear 
		\item \underline{\hspace{0.5in}}Type Large enough to be 
			readable 
		\item \underline{\hspace{0.5in}}Proper amount of 
			information per page (Not too much).
		\end{itemize}
	\item Questions and Answers
		\begin{itemize}
		\item \underline{\hspace{0.5in}}Were questions handled
			well by the speaker?
		\item \underline{\hspace{0.5in}}Did the questions show a
			lack of understanding by the audience?
		\item \underline{\hspace{0.5in}}Did the questions and
			answers clear up some of your comprehension of 
			the topic?
		\end{itemize}
\newpage
	\item Summary
		\begin{itemize}
		\item \underline{\hspace{0.5in}}Do you understand the
			big picture of the topic presented?
		\item \underline{\hspace{0.5in}}Did you get lost in some
			of the details?
		\end{itemize}
		\vspace{0.5in}
	\item \underline{\hspace{0.5in}}\% What percentage grade 
			would you give the presentation? \\
			{\bf (Note: this should be between 0 and 100)}
		
	\item Comments:
		\begin{itemize}
		\item .\hrulefill .
		\item .\hrulefill .
		\item .\hrulefill .
		\item .\hrulefill .
		\item .\hrulefill .
		\item .\hrulefill .
		\item .\hrulefill .
		\item .\hrulefill .
		\item .\hrulefill .
		\item .\hrulefill .
		\item .\hrulefill .
		\item .\hrulefill .
		\item .\hrulefill .
		\item .\hrulefill .
		\item .\hrulefill .
		\item .\hrulefill .
		\item .\hrulefill .
		\item .\hrulefill .
		\item .\hrulefill .
		\item .\hrulefill .
		\item .\hrulefill .
		\item .\hrulefill .
		\end{itemize}
	\end{itemize}
\end{document}


